Author's response to reviews

Title: Autonomic neuropathy in Fabry disease: a prospective study using the Autonomic Symptom Profile and cardiovascular autonomic function tests

Authors:

Marieke Biegstraaten (M.Biegstraaten@amc.uva.nl)
Ivo N. van Schaik (i.n.vanschaik@amc.uva.nl)
Wouter Wieling (w.wieling@amc.uva.nl)
Frits A. Wijburg (f.a.wijburg@amc.uva.nl)
Carla E.M. Hollak (c.e.hollak@amc.uva.nl)

Version: 4 Date: 1 June 2010

Author's response to reviews:

The BMC Series Editorial Production Team

Marieke Biegstraaten
Academic Medical Center
Dept of Neurology
PO box 22660
1100 DD Amsterdam, The Netherlands
T: +31 20 5666642 F: +31 20 6917682 E: m.biegstraaten@amc.uva.nl

June, 1, 2010

Dear Editorial Production Team,

Please find enclosed a revised version of our manuscript entitled “Autonomic neuropathy in Fabry disease: a prospective study using the Autonomic Symptom Profile and cardiovascular autonomic function tests.” by M. Biegstraaten, I.N. van Schaik, W. Wieling, F.A. Wijburg and C.E.M. Hollak.

We have revised our manuscript in accordance with your requests. A point-by-point description of the changes made is attached.

We are looking forward to hearing from you.

Yours sincerely,

Marieke Biegstraaten
List of contents included in the submission:

Article including figure legend and Tables 1, 2, 3 and 4 with changes marked

Figure 1

Point-by-point description of the changes made:

1. The figure legend contains graphics. These will either need to be included within the image as a key, or written in text (e.g. "black square").

   Answer: The graphics have been changed into written text

2. Tables: table 2 has been divided into parts, a, b and c. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

   Answer: The tables as well as the references to them in the text have been renumbered

3. Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system.

   Answer: Done

4. Tables- please remove the visible vertical lines from your tables

   Answer: Done

5. Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

   Answer: We should appreciate it greatly if you can make this change for us

6. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

   Answer: Done